Author’s response to reviews

Title: Clinical application of invalid foods using mealworms and evaluation of nutrition status and immune function: A study protocol for a randomized, double blind, placebo-controlled trial

Authors:

Hyung Sun Kim (milk8508@yuhs.ac)
Yun Sun Lee (GOLYS21@yuhs.ac)
Soo Yun Jang (helen0223@hanmail.net)
So Young Jun (SSONG@yuhs.ac)
Jin Hong Lim (DOCTORJIN@yuhs.ac)
Im Kyung Kim (ELLE0119@yuhs.ac)
Hyung Mi Kim (HMKIMQU@yuhs.ac)
JOON SEONG PARK (jspark330@yuhs.ac)

Version: 1 Date: 28 Jul 2019

Author’s response to reviews:

1. As your study has not been conducted yet, please rewrite the statement regarding participant consent so that it is presented in implicated future tense:

“Participants will provide written informed consent for participation.”

Author: We rewrite as review comment

2. As your study has not been conducted yet, please rewrite the statement regarding Availability of data and materials so that it is presented in implicated future tense:

“Data sharing is not applicable to this article as no datasets have been generated or analysed for this study yet.

Author: We rewrite as review comment
3. In your Funding section, please also state the role of the funding body in the design of the study; collection, analysis, and interpretation of data; and in writing the manuscript.

Author: We added the statements as review comment

4. Please relocate the figure legend section so that it is presented after the References.

Author: We relocated as review comment

5. If two authors have the same initials, please distinguish between them by designating them with numbers 1 and 2. For example: SYJ1 and SYJ2. SYJ1 would correspond to the author furthest up on the author list.

Author: We distinguish two authors as review comment

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Authors: We did